Author's response to reviews

Title: A biochemical marker panel in MRI-proven hyperacute ischemic stroke - a prospective study

Authors:

Carolin Knauer (karolin.knauer@uni-ulm.de)
Katharina Knauer (katharina.knauer@uni-ulm.de)
Susanne Müller (susanne.mueller@rku.de)
Albert C Ludolph (albert.ludolph@rku.de)
Dietmar Bengel (dietmar.bengel@oberschwabenklinik.de)
Hans-Peter Müller (hans-peter.mueller@uni-ulm.de)
Roman Huber (r.huber@uni-ulm.de)

Version: 2 Date: 12 September 2011

Author's response to reviews:

The following changes where made:

1. The methods chapter was completed with a declaration of given written informed consent for participation and the approval of the local ethics committee.

2. At the end of the manuscript following chapters where completed: List of abbreviations, competing interests, authors' contributions.